From: Mo Zheng
Department of Pathology,
Tianjin Fifth Central Hospital,
No.41 Zhejiang Road, Binhai New Area, Tianjin 300450, China.
Tel: (86) 13920114280
zhengzz1978@126.com
Nov 11, 2021
To: Editor

World Journal of Clinical Cases

Dear editorial board of World Journal of Clinical Cases,

We are very excited to receive the emails from you for the revising of our manuscript (Manuscript NO:71715, Case Report), firstly we want to express our great thanks for your great help and the acceptance of our case.

Following the requirements, we have revised the manuscript, and answered the questions point-to-point below.

To reviewer 1:
We revised the content in the case presentation for the specific comments with color markers.
1 The patient had no chief complain as the tumor was found quite incidentally by a routine physical examination.
2 The patient also had no history of present complains.
3 For the operative challenges, we add some contents in the paragraph of TREATMENT, as there was no adhesion between the kidney and other organs, the operation process was smooth. Only the choice of a proper surgical program was discussed preoperation.

To reviewer 2:
We have found a professional English language editing company to improved English.

To science editor:
1 The tumor was found quite accidently by a routine physical examination organized by the community
hospital every year. And even CT revealed a tumor in the left kidney, there was still no vital sign in following the physical exam, and we have listed the negative signs in the paragraph of physical examination.

2 we have found the right CT image of the case. A tumor was shown in the left kidney with a red arrow.

3 For the impact on the management of the patients, we add some contents in the paragraph of discussion with color markers about the adjuvant therapy.

To company editor-in-chief:
According to the comments and criteria, we have revised the manuscript. And we also have prepared the Signed Informed Consent Form and the documents of treatment in Chinese, the original figure document in a PowerPoint, all of them will be uploaded.

Thanks very much for your attention and your great help for our paper and us. If we have any mistake, we hope you can give us another chance to correct it, and we will do our best to finish the submitting of the revised manuscript.

Best wishes.

Sincerely yours,

Jing Yin and Mo Zheng.